Author's response to reviews

**Title:** Severe and aggressive cognitive SCA17 phenotype in a family with only marginally expanded CAG/CAA repeats in the TATA-box Binding Protein gene: A case report

**Authors:**

Troels T Nielsen (troelsn@sund.ku.dk)
Skirmante Mardosiene (smar0003@bbh.regionh.dk)
Anne Mette Løkkegaard (alokkegaard@dadinet.dk)
Jette Stokholm (jette.stokholm@rh.hosp.dk)
Susanne Ehrenfels (ssti0002@bbh.regionh.dk)
Lars Friberg (lars@fribergs.dk)
Jens K Nielsen (jnie0016@bbh.regionh.dk)
Jørgen E Nielsen (jnielsen@sund.ku.dk)

**Version:** 2 **Date:** 4 January 2012

**Author's response to reviews:** see over
Dear Editor-in-Chief

Please find attached a revised version of our manuscript “Severe and aggressive cognitive SCA17 phenotype in a family with only marginally expanded CAG/CAA repeats in the TATA-box Binding Protein gene: A case report”. We have changed the consent statement according to your requirements in order to be able to proceed to the review phase. Furthermore, I have forwarded you a copy of the consents by e-mail as requested.

Kind regards

Troels Tolstrup Nielsen
Department of Cellular and Molecular Medicine, Section of Neurogenetics
The Panum Institute, University of Copenhagen
Blegdamsvej 3
DK-2200, Copenhagen, Denmark,

Phone: +45 35 32 78 30, Fax: +45 35 32 78 45
troelsn@sund.ku.dk